Author's response to reviews

Title: Duration and adherence to antidepressant treatment in immigrant and native-born populations in Spain: a four year follow-up descriptive study.

Authors:

Inés Cruz (icruz lleida ics gencat cat)
Catalina Serna (cserna p lleida ics gencat cat)
Montserrat Rué (montse rue cmb udl cat)
Jordi Real (jreal lleida ics gencat cat)
Jorge Soler-Gonzalez (jorgesolergonzalez@gmail.com)
Leonardo Galván (lgalvan catsalut cat)

Version: 3 Date: 2 November 2011

Author's response to reviews:

Dear Editor,

we apologize for our mistake, there was indeed a missing citation (nº15) which refers to a previous publication of the authors in BMC Public Health. We have corrected it and we are uploading the new version.

Thanks for your understanding.